Author's response to reviews

Title: Reporting of harm in randomized controlled trials evaluating stents for percutaneous coronary intervention

Authors:

Morgane Ethgen (morganeethgen@hotmail.com)
Isabelle Boutron (isabelle.boutron@bch.ap-hop-paris.fr)
Philippe Gabriel Steg (gabriel.steg@bch.aphp.fr)
Carine Roy (carine.roy@bch.ap-hop-paris.fr)
Philippe Ravaud (philippe.ravaud@bch.ap-hop-paris.fr)

Version: 8 Date: 25 March 2009

Author's response to reviews:

As requested, we highlighted all changes (written in red) in the manuscript. We updated the analysis to include all trials published after 2005. Only numbers were updated, the text of the manuscript remains unchanged.

As requested, we added a conclusion section.